Author's response to reviews

Title: Fetal ERAP2 variation is associated with preeclampsia in African Americans in a case-control study

Authors:

Lori D Hill (hillld4@vcu.edu)
DaShaunda D Hilliard (ddh9@hermes.hood.edu)
Timothy P York (tpyork@vcu.edu)
Sindhu Srinivas (ssrinivas@obgyn.upenn.edu)
Juan P Kusanovic (jkusanov@med.wayne.edu)
Ricardo Gomez (rg.cedip@gmail.com)
Michal A Elovitz (melovitz@obgyn.upenn.edu)
Roberto Romero (prbchiefstaff@med.wayne.edu)
Jerome F Strauss III (jfstrauss@vcu.edu)

Version: 4 Date: 6 May 2011

Author's response to reviews: see over
April 20, 2011

Dear Editors,

Thank you for your consideration of our recent submission: MS:1274824451503480, “Fetal ERAP2 variation is associated with preeclampsia in African Americans in a case-control study.” Please find below our detailed responses to the formatting changes requested.

Sincerely,

Jerome F. Strauss III, M.D., Ph.D.

I. Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

Response: This has been added as requested.

II. coloured text /tracking: Please remove all coloured text and tracking from the manuscript.

Response: All colored text and tracking has been removed from the manuscript.

III. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

Response: The manuscript has been reviewed and represents the final version for publication.